Author’s response to reviews

Title: Phytochemical investigation and nephroprotective potential of Sida cordata in rat

Authors:

NASEER SHAH (drnaseeralishah@gmail.com)
Muhammad Khan (mrkhanqau@yahoo.com)
Dereje Nigussie (dere_nig@hotmail.com)

Version: 3 Date: 11 Jun 2017

Author’s response to reviews:

The manuscript has been greatly improved and only few minor corrections needed as shown in the attached documents.

All the recommendation were followed.

For the histology figure: it would look much better if the authors could point to the changes on the figure by using arrows, and label each part of the figure by Figure 1a, Figure 1b, ...

Figures are labelled as 1a,1b,1c,........But no change is made to the labeling of changes on the figures because we have lost the original one and now it looks difficult to remove it from the submitted ones and paste it outside.